
\documentclass{pj}
\usepackage{graphicx}


\begin{document}
\setcounter{page}{1}
\pjheader{Vol.\ x, y--z, 2004}
%\jheader{Vol.\ x, y--z, 2004}
%\pheader{Vol.\ x, y--z, 2004}

\title{Replace this line with the title of your manuscript}

\author{F.~M.~LastName and F.~M.~LastName}

\address{Author Affiliation or Institute\\
Address, format is like 77 Massachusetts Ave. 26-305, MA 02139, USA
}  

\author{F.~M.~LastName}

\address{Author Affiliation or Institute\\
Address
}  


\runningauthor{Corresponding Author's Last Name}
\tocauthor{Initial of First Name.~Last Name}

\begin{abstract}
The abstract of your article should be here. 
\end{abstract}

%
%\mytableofcontents
\tableofcontents

\section{Your Section Heading Should Be Here.}
\label{section label}

The heading of all the sections and subsections should have 
every word begin with the capital letters.

This space is for the contents. The section should be labeled by using 
$\backslash$label\{ section label\}. To refer this section, use 
$\backslash$ref\{section label\}.


\section{Formulations and Equations}
\label{sec:formulation}

For equations, 
\begin{equation}
\nabla \times \overline E(\overline r,t)=-\frac{\partial \overline B(\overline r,t)}{\partial t}
\end{equation}

For convenience, authors are allowed to use the \\ newcommand to define a 
short name for the operators, but please define the newcommand at the beginning
 this main file only.

For multiline equations, authors can use "eqnarray"

\begin{eqnarray}
\nabla \times \overline H(\overline r,t)=\frac{\partial \overline D(\overline r,t)}{\partial t}+\overline J(\overline r,t)  \nonumber \\
\nabla \cdot \overline B(\overline r,t) =0
\label{eq:equation label}
\end{eqnarray}

The equation is referred as (\ref{eq:equation label}).

\section{Figures and Tables}
\label{sec:FiguresTables}

\subsection{Figures}
For best publishing quality, we recommend the EPS format for figures.
Fig.~\ref{fig:test} shows how to insert an encapsulated postscript figure.

\begin{figure}[h]
\centerline{\includegraphics[width=0.8\columnwidth,draft=false]{fig1.eps}}
\caption{This is a sample figure.}
\label{fig:test}
\end{figure}

\subsection{Tables}

Table~\ref{tab1} shows how to display a table.

\begin{table}[h]
\renewcommand{\arraystretch}{1.3}
\caption{A Sample Table. }
\vskip0.2in
\begin{center}
\small \begin{tabular}{|c|c|c|}\hline
col 1's label & col 2's label & col 3's label  \\ \hline
1 & 2 & 3  \\ \hline
130 &0.5 & 1.159 \\\hline
  \end{tabular}
\end{center}
\label{tab1}
\end{table}

\section{Citations}

The references are cited as \cite{article,book}.

\ack

The acknowledgment should be here.

\appendixx{This is the First Appendix}

\subsection{Sub-Appendix}

\begin{equation}
E=mc^2
\end{equation}

\appendixx{This is the Second Appendix}

\begin{equation}
F=ma
\end{equation}



\begin{thebibliography}{99}

\bibitem{article} Lastname,~F.~M., ``Title of the journal paper," {\it Journal Title Abbreviation\/}, Vol.~34, No.~10, 1064--1076, 1986.

\bibitem{book} Lastname,~F.~M., {\it Book Title \/}, Wiley-Interscience, 
New York, 1986.

\end{thebibliography}

\begin{biography}
{\bf First Name Middle Initial.\ Last Name} A short biography for the author.

\end{biography}

\end{document}
